Author's response to reviews

Title: Association of Nrf2-encoding NFE2L2 haplotypes with Parkinson's disease

Authors:

Malin von Otter (malin.vonotter@gu.se)
Sara Landgren (sara.landgren@pharm.gu.se)
Staffan Nilsson (staffan@chalmers.se)
Dragana Celojevic (dragana.celojevic@gu.se)
Petra Bergström (petra.bergstrom@clinchem.gu.se)
Anna Håkansson (anna.hakansson@pharm.gu.se)
Hans Nissbrandt (hans.nissbrandt@pharm.gu.se)
Marek Drozdzik (drozdzik@sci.pam.szczecin.pl)
Monika Bialecka (monika-bialecka@post.pl)
Mateusz Kurzawski (mkurz@op.pl)
Kaj Blennow (kaj.blennow@neuro.gu.se)
Michael Nilsson (michael.nilsson@vgregion.se)
Ola Hammarsten (ola.hammarsten@clinchem.gu.se)
Henrik Zetterberg (henrik.zetterberg@clinchem.gu.se)

Version: 6 Date: 25 February 2010

Author's response to reviews: see over
To whom it might concern

On your request I have checked the manuscript for any changes that I would like to do after your formatting. I have made the following changes.

1. I have made sure that there are no “double line breaks” before any headings and that the headings are on the same page as the connected paragraph. There were some mistakes of this kind in the formatted version you returned to me.

2. I have changed the tables in order to make sure that the text has not been divided into two rows in any of the cells which was the case in some of the cells in the formatted version I received from you. I have not changed the size of the font, only changed the width of the columns, and in the case of table 3 I have removed the first column and included the gene name in the SNP column in the same way as in table 2.

3. I have included a new version of the picture since it had been wrongly cropped. I would like the picture fully framed. In the formatted version you sent me back the left part of the frame was missing.

Yours sincerely,

Malin von Otter, PhD

Institute of Neuroscience and Physiology
Department of Psychiatry and Neurochemistry
The Sahlgrenska Academy at University of Gothenburg
S-413 45 Gothenburg, Sweden